Author's response to reviews

Title: BACH1 Ser919Pro variant and breast cancer risk

Authors:

Pia Vahteristo (pia.vahteristo@helsinki.fi)
Kristiina Yliannala (kristiina.yliannala@ktl.fi)
Anitta Tamminen (anitta.tamminen@hus.fi)
Hannaleena Eerola (hannaleena.eerola@fimnet.fi)
Carl Blomqvist (carl.blomqvist@hus.fi)
Heli Nevanlinna (heli.nevanlinna@hus.fi)

Version: Date: 5 December 2005

Dear Dr. Puebla, Helsinki, Dec 5th 2005

Please find enclosed the revised manuscript "BACH1 Ser919Pro and breast cancer risk" (MS: 1497578677791403). All the suggested changes have been made:

*Red highlighting referring to the changes made after reviewer's comments has been removed.
*Patients and Methods section has been renamed into Methods.
*Reference list has been modified such that all the author names are listed.
*Table 2 has been modified as suggested.
*Title for table 3 has been provided.

Manuscript formatting checklist has also been checked. We hope that the manuscript now fulfils the high standard of your journal and could be accepted for publication.

Yours sincerely,

Pia Vahteristo, PhD
Department of Medical Genetics
Biomedicum Helsinki, Rm B520a
PO Box 63 (Haartmaninkatu 8)
FIN-00014 University of Helsinki
Finland
Tel: INT+358-9-19125600, Fax: INT+358-9-19125105
email: pia.vahteristo@helsinki.fi